Author's response to reviews

Title: Schistosomiasis transmission at high altitude crater lakes in Western Uganda

Authors:

John Rubaihayo (rubaihayoj@yahoo.co.uk)
Ezekiel Mughusu (bmughusuezkiel@yahoo.com)
Philbert Clouds (closim@yahoo.com)

Version: 2 Date: 20 February 2008

Author's response to reviews:

Dear Editor BMC Infectious Diseases,

I have included the section on ethical considerations and more recommendations in the discussion.

Hope its now ok

John